Author’s response to reviews

Title: Effect of 6-months of physical exercise on the nitrate/nitrite levels in hypertensive postmenopausal women

Authors:

Pedro R Zaros (przaros@uol.com.br)
Carla EMR Pires (carlapires@hotmail.com)
Mauricio Bacci Jr (mbacci@rc.unesp.br)
Camila Moraes (mormila@hotmail.com)
Angelina Zanesco (azanesco@rc.unesp.br)

Version: 3 Date: 17 June 2009

Author’s response to reviews:

June 17th, 2009

Dear Dr Melissa Norton
Editor-in-chief of BMC journal

Please find attached the revised manuscript entitled: Effect of 6-months of physical exercise on the nitrate/nitrite levels in hypertensive postmenopausal women by Zaros et al.

In response to major revisions:
Acknowledgements was inserted in the text
minor revision:
the envelope symbol was removed
the intire text was revised and some erros were removed

My best regards

Angelina Zanesco
associate professor
UNESP